\begin{table}%t1
\caption{Properties of blinkers and explosive events. \label{blink_ee_prop}}
%\centerline
{
\begin{tabular}{lcc}
\hline\hline
&QS & Explosive \\
Property & blinkers & events\\
\hline\\[-3mm]
Global frequency (s$^{-1}$)& 7.5$^1$ & 44$^3$, $\approx$600$^4$\\
Mean intensity \% increase & 80$^1$ & 60$^5$\\
Mean area (Mm$^{2}$)& 29$^1$ & $\approx$2.0$^4$\\
Mean lifetime (min)& 16.4$^1$ & 1.0$^4$\\
Maximum Line Shift (\kms) & 30$^2$ & 110$^4$\\
Maximum Line Width (\kms) & 45$^2$ & -- \\
\hline
\end{tabular}}
\medskip
$^1$ \citet{Bewsher2002}.\\
$^2$ \citet{Bewsher2003}.\\
$^3$ \citet{Cook1987}.\\
$^4$ \citet{Dere1989}.\\
$^5$ \citet{Brkovic2004}.
\end{table}